Author’s response to reviews

Title: Rapid decrease in carotid plaque lipid content as assessed by magnetic resonance imaging during treatment of rosuvastatin

Authors:

Rui-Xue Du (drx1972@hotmail.com)
Jian-Ming Cai (beili12345@sina.cn)
Xue-Qiao Zhao (xzhaocardiology.washington.edu)
Qing-Jun Wang (wangqijun77@163.com)
Dan-Qing Liu (ldq562892650@126.com)
Wen-Xiu Leng (lenslin301hosp@yahoo.com)
Peng Gao (jiandangaop@126.com)
Hong-Mei Wu (wuhi301@sina.com)
Ping Ye (yeping@sina.com)
Lin Ma (cjr.malin@vip.163.com)

Version: 2 Date: 25 January 2014

Author’s response to reviews: see over
Dear Prof. Erica Cruz,

Thank you for giving very professional advice to our paper. According to your suggestions, I have made some revision in part of competing interests.

Please contact me if you have any questions or need any additional information. Once again, thank you for your help to our paper processing.

Yours sincerely,

Ping Ye, MD.

E-mail: yeping301@sina.com